Author's response to reviews

Title: Long-term cancer mortality time trends in Ohio, USA, 1970-2001: differences by race, gender and age

Authors:

Jerzy E Tyczynski (jerzytycz@yahoo.com)
Hans (J) Berkel (berkel@cancerpreventioninstitute.org)

Version: 7 Date: 12 October 2005

Author's response to reviews:

Tables 1 & 2 have been included into the manuscript (at the end) in the form of Word tables. I hope this time it would be ok.
Jerzy Tyczynski